Author’s response to reviews

Title: Histone deacetylase inhibition by Entinostat for the prevention of electrical and structural remodeling in heart failure

Authors:

Johanna Freundt (j_freu03@uni-muenster.de)

Gerrit Frommeyer (Gerrit.Frommeyer@ukmuenster.de)

Tilmann Spieker (Tilmann.Spieker@ukmuenster.de)

Fabian Wötzel (woetzel@pathologie-muenster.de)

Jochen Schulze Grotthoff (Jochen.SchulzeGrotthoff@ukmuenster.de)

Jörg Stypmann (Stypmann@ukmuenster.de)

Georg Hempel (hempege@uni-muenster.de)

Michael Schäfers (schafmi@uni-muenster.de)

Andreas Jacobs (ahjacobs@uni-muenster.de)

Lars Eckardt (lars.eckardt@ukmuenster.de)

Philipp Sebastian Lange (PhilippSebastian.Lange@ukmuenster.de)

Version: 2 Date: 06 Dec 2018

Author’s response to reviews:

Dear Professor Wainford, dear Editorial team,

Thank you very much again for re-reviewing our manuscript and considering it for publication. On behalf of my co-authors I would like to resubmit the above named manuscript to "BMC Pharmacology & Toxicology". Following the advice of your reviewers, we have carefully revised the manuscript. We hope that you and your reviewers will now find the manuscript appropriate for publication in your journal. Below please find our point-by-point answers to the reviewers’ comments.
Authors had made substantial changes in the text of manuscript to address previous concerns. Here are some of minor concerns:

1) All western blots images in the manuscript need to be quantified.

Response: All Western Blots in the manuscript were quantified. The additional data was added to supplemental figure 3. The figure legend was updated accordingly.

2) Most of manuscript images and newly added text have comma (,) for indicating decimal point instead period(.). I would recommend authors to change it.

Response: All commas indicating decimal points were corrected accordingly.